Author's response to reviews

Title: Infection by H1N1 pandemic virus and Tnf 308, Tnf 252, Tnf 238, Tnf 376, Il1Beta, IL6, II8, ccl1 in Mexican population: Case-Control Study

Authors:

Guadalupe Morales-García (lumorga@yahoo.com.mx)
Ramcés Falfán-Valencia (dcb_rfalfanv@hotmail.com)
Román Alejandro García-Ramírez (garra22noviembre@yahoo.com.mx)
Angel Camarena (ang_edco@yahoo.com.mx)
Alejandra Ramírez-Venegas (aleravas@hotmail.com)
Manuel Castillejos-López (mcastillejos@gmail.com)
Martha Pérez-Rodríguez (meperezrodriguez@yahoo.com.mx)
César González-Bonilla (cesar.gonzalezb@imss.gob.mx)
Concepción Grajales-Muñiz (concepcion.grajales@imss.gob.mx)
Víctor Borja-Aburto (victor.borja@imss.gob.mx)
Juan Manuel Mejía-Arangure (juan.mejiaaa@imss.gob.mx)

Version: 2 Date: 5 March 2012

Author's response to reviews:

Diana Marshall PhD, BMC Infectious Diseases

Dear Dr. Marshall,

We submit our manuscript, entitled "TNF -308, TNF +252, TNF -238, TNF -376, IL1#, IL6, IL8, CCL1 AND INFECTION BY H1N1 PANDEMIC VIRUS ", for consideration for publication in BMC Infectious Diseases by Morales-García et al.

We review our manuscript and add the next statement in methods section: This study was approved by the Ethics Board of the National Institute of Respiratory Diseases (INER B05-10). The written informed-consent forms were legible for cases and controls, and the information was obtained through the use of clinical records according to the Mexican Official Standard NOM-168-SSA1-1998.

We thank you in advance for your time in considering this article for publication in BMC Infectious Diseases.

Sincerely,

Dr. Juan Manuel Mejía-Aranguré
Senior Researcher
Clinical Epidemiology Research Unit, High Specialty Medical Unit, Pediatric Hospital, National Medical Center "Century XXI"
National Commission of Scientific Research of the Mexican Institute of Social
Security
Juan.mejiaa@imss.gob.mx ; arangurejm@hotmail.com